Author's response to reviews

Title: Referral Patterns and Attitudes of Primary Care Physicians Towards Chiropractors

Authors:

Barry R Greene Dr (barry-greene@uiowa.edu)
Monica Smith Dr (smith_m@palmer.edu)
Veerasathpurush Allareddy (vallared@mail.public-health.uiowa.edu)
Mitch Hass (mhaas@wschiro.edu)

Version: 2 Date: 28 December 2005

Author's response to reviews: see over
December 28, 2005

Dr. Peter Newmark, Editor-in-Chief
BioMed Central Ltd
Current Science Group
Middlesex House
34-42 Cleveland Street
London W1T 4LB
United Kingdom.

Dear Drs. Peter Newmark and Betsy B. Singh,

Thank you for the kind review of our manuscript titled “Referral Patterns and Attitudes of Primary Care Physicians Towards Chiropractors” and giving us the opportunity to further revise and improve our manuscript. We have taken the reviewers comments into consideration and modified it as best as possible. We have discussed about the issue of response rate in our manuscript and have cited the references as suggested by the reviewer. We have also cited the article “Interprofessional Referral Patterns in an Integrated Medical System”. We hope that you will now find this manuscript acceptable for publication.

Sincerely,

Barry R. Greene, Ph.D.
Professor and Interim Head of Department
Dept Health Management & Policy
College of Public Health
200 Hawkins Drive
E212 GH
Iowa City, Iowa
52242-1008
barry-greene@uiowa.edu
Phone 319 384 5135
Fax 319 384 5125